Author’s response to reviews

Title: Septicemia, necrotizing fasciitis, and peritonitis due to Vibrio vulnificus treated with early use of polymyxin B hemoperfusion in a patient undergoing CAPD: A case report

Authors:

Jae Young Kim (kim.jy@yuhs.ac)
Young Su Joo (jys8427@gmail.com)
Lee Sang Mi (leesm2332@yuhs.ac)
Ji Young Lee (jy0330@yuhs.ac)
Jung Tak Park (jtpark@yuhs.ac)
Seong Hyeok Han (hansh@yuhs.ac)
Tae-Hyun Yoo (yoosy0316@yuhs.ac)
Shin-Wook Kang (kswkidney@yuhs.ac)

Version: 3 Date: 17 Mar 2020

Author’s response to reviews:

March 15th, 2020

Doris Cuckovic, DDS, PhD
Editor in Chief
BMC nephrology

Re: Decision letter of manuscript ID: BNEP-D-19-00525, entitled “Septicemia, necrotizing fasciitis, and peritonitis due to Vibrio vulnificus treated with early use of polymyxin B hemoperfusion in a patient undergoing CAPD: A case report”
Dear Dr. Cuckovic,

Thank you very much for reviewing our manuscript and for your email (sent on March 13th) encouraging us to resubmit a revised version for publication in BMC nephrology.

We are pleased to submit our manuscript, which has been revised and updated again according to the reviewer’s second suggestions and comments. Below you will find our detailed, point-by-point responses to each comment from the reviewer.

Thank you again for your time and effort in dealing with our manuscript. We sincerely hope that the revised manuscript will meet the standard for publication in BMC nephrology.

We look forward to hearing back from you.

Sincerely,

Shin-Wook Kang, M.D., Ph.D.
Yonsei University, Institute of Kidney Disease Research
College of Medicine, Department of Internal Medicine
134 Sinchon-dong, Seodaemun-gu, Seoul 120-752, Republic of Korea
Phone: +82-2-2228-2281
Fax: +82-2-393-6884
E-mail: kswkidney@yuhs.ac

Responses to Editor’s comments

Comments to the Authors: Your manuscript "Septicemia, necrotizing fasciitis, and peritonitis due to Vibrio vulnificus treated with early use of polymyxin B hemoperfusion in a patient undergoing CAPD: A case report" (BNEP-D-19-00525R2) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Nephrology, once you have carried out some essential revisions suggested below.
Answer: The authors appreciate the editor for the comments. Detailed responses for each comment have been addressed below.

Comment 1: We note that LSM is included in the authors list, but is not included in the Authors Contributions section. Similarly, we note that SML is included in the Authors Contributions but is not included in the Authors list. As such, please clarify whether this is the same author and amend the initials. Please note that the first initial should be the Authors first name (given name) and the last initial should be the authors last name (family name).

Answer: Sangmi Lee in the authors list is the same author who is included in the Authors Contributions section as SML. We corrected the name of the author which was filled mistakenly.

Comment 2: Please remove the consent section that was included before the declarations and only include the consent for publication statement, in the Consent for Publication section.

Answer: The consent section before declarations is removed and included in the Consent for Publication section.

Comment 3: If you do not wish to publish the CARE checklist then please remove it from the file inventory and include a statement within the manuscript that you adhered to CARE guidelines/methodology.

Answer: CARE checklist has been removed from file inventory. The statement that the authors adhered to CARE guidelines/methodology has been included in the revised manuscript.

Comment 4: Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Answer: The final form for publication of the revised submission has been uploaded.